Dear editor,

On behalf of all the contributing authors, I would like to express our sincere appreciations of your letter and reviewers’ constructive comments concerning our article entitled “Comparison of drug concentrations in blood and gastric lavage fluid before and after gastric lavage: A case report” ((Manuscript NO.: 87722, Case Report). These comments are all valuable and helpful for improving our article. Based on the reviewers' comments, we have revised the manuscript and highlighted the revised/added contents with yellow color in the revised manuscript

Answer to reviewer:
Reviewer 1 Dear author, thank you for sharing your research. It is a bold and fascinating endeavor, still in its early stages, that deserves to be expanded with new cases and substances. I believe that within the limitations of your study, one must consider the varying absorption times of substances at the gastric level. Regards
Response: We feel great thanks for you professional review work on our report. As you point out our research is still in its infancy and we are planning further studies to determine the clearance efficiency of gastric lavage at different times after ingestion of different substances in order to better guide the clinical development of treatment protocols.
Reviewer 2 It was a nicely written article. It would be better if you provide recommendation for further research and add a conclusion section. If you can add few other references mentioning interval between ingestion and starting of GL, it would be better.
Response: We sincerely appreciate the valuable comments. We have added a conclusion section at the end of the article as well as next steps for research. We checked the literature carefully and added more reference about interval between ingestion and starting of GL. Thank you again for your positive comments and valuable suggestion to improve the quality of our manuscript.